Author’s response to reviews

Title: Myocardial Infarction during anaphylaxis in a young healthy male with normal coronary arteries- Is epinephrine the culprit?

Authors:
withanage jayamali (dhanushka.withanage1987@gmail.com)
tharuka herath (tharukaherath11@gmail.com)
aruna kulathunga (arunakulathunga@yahoo.com)

Version: 2 Date: 26 Jul 2017

Author’s response to reviews:

Reviewer reports:

Manan Pareek, M.D., Ph.D. (Reviewer 1): The manuscript has definitely improved. The flow is much better. Few errors remain, but can be corrected quickly.

In addition, please check the verbs throughout the manuscript, including the abstract, and please capitalize "Kounis".

I capitalized the “Kounis” word and corrected grammar, verbs and other errors that were there in the manuscript.